NEETs in Latin America and the Caribbean: Skills, Aspirations, and Information

Alfredo Alvarado, Belén Conde, Rafael Novella^{1,2} and Andrea Repetto^{3*}

Abstract

This paper studies the role of cognitive skills, socioemotional skills, aspirations, and expectations on the likelihood of being NEET (not in education, training or employment) using a novel sample of 15 to 24 year old residents of seven Latin American and Caribbean countries. After controlling for sociodemographic household conditions, we find that numeracy and literacy skills, core self evaluation, extraversion and educational aspirations are robust correlates of being NEET. We also find cross country heterogeneity. That is, in some countries passion and perserverance for long term goals, neuroticism, and labor market information biases are additional factors associated with being NEET.

Keywords: cognitive skills, socioemotional skills, aspirations, expectations, information bias, youth

1. Introduction

About 20 million youths in Latin America and the Caribbean (LAC) are not in education, employment or training (NEET).⁴ That is, 21% of the population between the ages of 15 and 24 is NEET in LAC, which falls between the rates of North America (14%) and Europe and Central Asia (14%) and those of Asia (30%) and the Middle East and North Africa (32%).⁵ The NEET phenomenon is a global concern as it represents the underemployment of human

¹ Labor Markets and Social Security, Inter-American Development Bank, Washington, DC, USA.

² Oxford Department of International Development, The Centre on Skills, Knowledge and Organisational Performance, University of Oxford, Oxford, UK.

³ School of Government, Universidad Adolfo Ibáñez, Santiago, Chile.

^{*} Correspondence to: Andrea Repetto, School of Government, Universidad Adolfo Ibáñez, Diagonal Las Torres 2640, Santiago, Chile. E-mail: andrea.repetto@uai.cl

⁴ De Hoyos et al. (2016).

⁵ World Bank (2019a).

capital and a loss of productive opportunities. It is also associated with the risk of social exclusion. It thus affects both individuals' welfare and national economic growth.

There is a wealth of research analyzing the educational and employment decisions of youths with a focus on their decisions to continue their educations or to enter the labor market (Edmonds, 2008; Baland and Robinson, 2000). A smaller but growing body of research has explored the correlates of simultaneous educational and labor market inactivity, that is the decision to neither study nor work nor engage in training. Most of these studies focus on socioeconomic conditions such as gender, household income, having dropped out of school, and fertility. Only a small number of papers analyze the role played by aspirations and expectations, and socioemotional skills. Moreover, much of this latter research has been done in developed countries. Thus, it is important to expand the knowledge about the role of skills, aspirations, and expectations in the context of developing countries.

This paper explores the conditions associated with the likelihood of being a NEET youth in LAC, defined as young people who are not in education, employment, or training (ILO, 2015). More specifically, in addition to traditional socioeconomic conditions, we analyze the role of cognitive and non cognitive skills, aspirations, and expected educational returns on the likelihood of being NEET. Our emphasis on these variables is based on previous studies that show that they are important predictors of educational attainment and labor market outcomes (Heckman et al., 2006). Moreover, there is a mounting body of evidence that suggests that expectations, socioemotional skills and behavior can be modified during adolescence through low cost interventions (Novella and Repetto, 2019).

We use a new dataset, the Millennials in Latin America and the Caribbean survey, for our purpose (Novella et al., 2018). The survey contains in depth information on over 12,000 youths aged 15 to 24 who are residents of urban areas in seven LAC countries (Brazil, Chile, Colombia, El Salvador, Haiti, Mexico and Paraguay). According to official statistics, the proportion of youths aged 15 to 24 years who are NEETs in these countries is large, ranging from 16% in Paraguay in the past decade, to 27% in El Salvador (Figure 1). The unique database was designed to collect information on a wide range of topics associated with their schooling and working decisions, and allows us to do a cross country comparison of the relevance of different variables on the probability of being NEET.

⁶ See the literature review in Section 2 below.

In order to uncover their distinct correlations with being NEET, we add the variables of interest in sets, one by one, to our empirical model. To compare with the existing evidence, our basic model includes socioeconomic variables (age, gender, years of education, teenage parenthood, household income, and household composition indicators). We then progressively add cognitive skills, socioemotional skills, educational aspirations, and finally, expectations of educational returns.

As in the previous literature, we find relevant statistical associations with socioeconomic variables: being female, age, teenage parenthood and young children at home are positively correlated with the likelihood of being a NEET youth, while household income and years of education are significatively and negatively correlated. These results remain practically unchanged as additional covariates are added to the empirical model.

Our main contribution, however, is to show that cognitive and non cognitive skills are negatively correlated with the probability of being NEET even after correcting for socioeconomic characteristics. Specifically, better performance in cognitive tests and a higher score in a core self evaluation index — consisting of self esteem, self efficacy and locus of control — are all associated with a lower chance of being NEET. Similarly, the extraversion item of the Big Five Inventory (Rammstedt and John, 2007) is empirically associated with a lower likelihood of being NEET. We find no significant association with the other items of the Big Five Inventory or with an index measuring passion and perseverance for long term goals (the Grit scale; see Duckworth and Quinn, 2009).

Also important, our results show that educational aspirations are negatively correlated with the likelihood of being NEET. However, we find no association between being NEET with the expected return to higher education and with biases in labor market information.

Furthermore, we find cross country heterogeneity in our estimation results. That is, for some countries, passion and perserverance for long term goals, neuroticism, and labor market information biases are additional factors associated with being NEET.

Our results are robust to the set of included variables. This suggests that cognitive skills, socioemotional skills, and aspirations are independently correlated with educational and employment activities. That is, these factors are not mediated through socioeconomic conditions such as teenage pregnancy, educational attainment, or household income.

This result, in turn, has implications for the design of policies aiming to reduce the rates of NEET youths beyond programs that alleviate poverty and reduce the prevalence of teen parenthood. In particular, interventions that impact cognitive and non cognitive skills and aspirations also have potential to be effective in addressing the NEET phenomenom.

It is worth highlighting that our paper provides a cross sectional analysis aimed at examining conditional correlations between the probability of being NEET and the relevant variables. That is, the results in this paper should not be interpreted in a causal way because of the restrictions imposed by the cross sectional nature of the Millennials in LAC dataset. Even so, this paper provides novel evidence on the dimensions related to youths' decisions that have so far not received much attention in the academic and policy agendas in LAC.

The remainder of the paper is organised as follows. Section 2 briefly reviews the literature that studies the decision of being NEET. Section 3 describes the Millennials in LAC survey and our estimation strategy. Section 4 presents the results and discusses potential policy interventions. Finally, Section 5 concludes.

2. NEETs: conceptual framework and literature review

The analytical framework of schooling, work and time allocation by children and youths helps explain why youths may find themselves as NEET at some point in time (Skoufias and Parker, 2001; Edmonds, 2008; Berhman et al., 2015). According to this literature, the decision maker – the parents, the family, or the youths themselves– maximize welfare by allocating time to different activities according to the expected net marginal benefit of each. In the context of education and labor market activities, the factors theoretically affecting this relative expected net marginal benefit can be classified into three broad categories: (i) characteristics of the youths themselves, (ii) characteristics of their households, and (iii) characteristics of the educational environment and the local labor market.

Youths' characteristics

According to the analytical framework, the ability to learn and reap the benefits of education is an important factor in determining youths' NEET status. This ability affects the returns from and the opportunity cost of studying instead of working. Cognitive skills such as

numeracy and literacy skills allow youths to solve problems and understand information (OECD, 2015a). Non cognitive skills such as motivation, self esteem and self efficacy allow people to organize and plan, to relate to others, and to commit to a task, among others (Kautz et al., 2014; Behrman et al., 2015). Educational attainment also matters as it summarizes prior investments in human capital and as it serves as a signal about the youths' abilities (Pallais, 2014).

Also at the theoretical level, aspirations and subjective expectations of earnings also affect the time allocation decision as they describe how youths perceive the trade off between work and study when they are in the process of making human capital decisions (Dominitz and Manski, 1996; Wiswall and Zafar, 2015).

Age also plays a role (Edmonds, 2008). First, the cost of studying and not working increases with age, as the opportunity cost rises along with the limits to educational access. Second, additional decisions are taken jointly as youths age, including economic independence, fertility and engagement in risky behaviors (for example, smoking, drug consumption, and having unprotected sex). Finally, some social conventions become more important with age (for example, becoming a mother).

In the same line, gender also matters due to norms and stereotypes, as well as a due to gender differences in labor market outcomes that affect the net expected return to human capital investments (Attanasio and Kaufman, 2014).

Household characteristics

According to the analytical literature, household characteristics such as socioeconomic status also impact the work or study decision: in addition to any direct costs of schooling, staying in education entails the cost of forgone current earnings (Schultz, 1960; Basu and Van, 1998; Baland and Robinson, 2000). Also, the demographic composition of the household matters since fewer adults or a larger number of household members needing care may put pressure on youths to earn income or limit their study possibilities (Edmonds, 2008).

Educational environment and local labor markets

Finally, the theoretical literature stresses that access to educational establishments, the compulsory nature of education, the availability of educational financing, the quality of the

educational system, as well as labor market conditions, regulations and policies also affect the expected net marginal benefit of different activities and thus the time allocation decision among young people (OECD, 2014; Behrman et al., 2015; Jongbloed and Giret, 2019).

The existing empirical research on the factors associated to being in the NEET status has analysed the role of these three sets of variables.

Research focusing on individual socioeconomic conditions and family background shows that the NEET status is negatively correlated with the household's economic conditions (Mascherini et al., 2012; Baron et al., 2016; Noh et al., 2017, Alcázar et al., 2019), parental education (Mascherini et al., 2012; Alfieri, 2015), and the number of employed individuals in the household (Székely and Karver, 2015; Susanli, 2016). In addition being female (Yates et al., 2011; Baron et al., 2016; Wickremeratne and Dunusinghe, 2018; Alcázar et al., 2019), having low educational attainment and/or dropped out of school (Mascherini et al., 2012; OECD, 2015b; De Hoyos, et al. 2016; Salvà-Mut et al., 2018), marriage (Zudina, 2017; Favara and Sánchez, 2018), and having children (Alcázar et al., 2019) are all positively associated with being a NEET. Interestingly, these results hold for both developed and developing countries.

On the contrary, the literature about the effects of a wider set of individual level variables, especially cognitive, non cognitive skills and information, is scarce and, to the best of our knowledge, mostly limited to the context of developed countries. This research has found that the probability of being a NEET is negatively associated with cognitive skills (OECD, 2015b; Jongbloed and Giret, 2019), aspirations and expectations (Crawford et al., 2011; Yates et al., 2011; Noh and Lee, 2017), and with self esteem, external locus of control, and other psychological traits (Baggio et al., 2015; Cornaglia et al., 2015; Mendolia and Walker, 2015; Goldman-Mellor et al., 2016; Ng-Knight and Schoon, 2017; Bernal et al., 2018).

Finally, there is a wealth of research analyzing the relevance of the educational environment and labor market conditions on the probability of being a NEET in countries at different development stages. This research finds that the quality of the educational system, the features of employment regulations, labor market conditions and the macroeconomic environment, have an impact on the incidence of youth out of school and out of work

(Mascherini et al., 2012; OECD, 2015b; Székely and Karver, 2015; Jongbloed and Giret, 2019).

In sum, given the availability of data, most research has focused on the traditional determinants of the probability of being NEET (that is, socioeconomic and demographic variables). The richness of our database allows us to understand the role of less studied aspects such as cognitive and non cognitive skills and beliefs, and to extend this research to less developed countries.

3. Materials and Methods

3.1. The Millennials in Latin America and the Caribbean Survey

Our analysis is based on the Millennials in LAC survey, a cross-sectional survey conducted in Brazil, Chile, Colombia, El Salvador, Haiti, Mexico, and Paraguay. The survey was designed to better understand the schooling and labor market decisions of youths. The survey was administered between 2017 and 2018 and includes information on youths aged 15 to 24 years living in urban areas. Using a stratified multistage sampling method, households were selected based on previous censuses in each country. The final stage of the sampling method consisted of randomly choosing a young person within the household. The main characteristics of the survey are summarised in Table 1.

The survey consists of two questionnaires. The first contains standard demographic and socioeconomic questions. It also gathers information on skills, expectations, and aspirations among other variables. The second questionnaire contains information about risky behaviors. To improve the response rate and the quality of the information, this second questionnaire was self administered (Tourangeau et al., 1997; Krumpal, 2013). Written consent was

⁻

⁷ The data used in this paper was collected as part of the "Millennials in Latin America and the Caribbean: to work or study?" project. The project, including the data collection, was funded by the International Development Research Centre (IDRC - Canada) and the Inter-American Development Bank (IDB). The seven countries in the study were selected with the aim of having a representative sample of similar countries within the sub-regions in Latin America and the Caribbean (that is, Andean countries, Southern Cone, Central America and the Caribbean). The final selection of countries is the result of combining the interests of the donors and the availability of local partners interested in doing research in youth's school and work decisions within the project's budget and time constraints. Unfortunately, due to budget constraints it was not possible to collect data in more countries in the region.

obtained, either from the participants themselves if they were 18 or older or their parents otherwise.

One novel factor is that the survey looks at socioemotional skills, which recent literature has found to be an important factor for labor market outcomes (Almlund et al., 2011). More specifically, it includes the Duckworth and Quinn (2009) Grit test that measures perseverance and passion for long term goals; the Big Five Inventory (BFI-10) test for personality traits that measures extraversion, emotional stability, agreeableness, conscientiousness, and openness to experience (Rammstedt and John, 2007); the Rosenberg self esteem test that measures the image that people have of themselves (Rosenberg, 1965); a test for self efficacy that measures people's beliefs about their own abilities to achieve goals (Schwarzer and Jerusalem, 1995); and a locus of control scale that measures the degree to which people believe they have control of their outcomes in life events compared to those events being controlled by outside forces (Rotter, 1966). The questionnaire also measures perceptions about wages.

Finally, the survey includes basic reading comprehension and numeracy tests. The reading comprehension test asks subjects to fill twelve blanks in a text using listed words. The numeracy test poses simple problems in which respondents need to divide and multiply to obtain the correct answers (for example, split a given amount of money between five individuals).

3.2. Study measures

We include traditional sociodemographic control variables such age and gender. Education is measured by the number of years of schooling achieved. We also include a teenage parenthood dummy that equals one if the youth had a child when younger than 20 years old or is pregnant and younger than 20, and zero otherwise. Since household's monthly income per capita is measured in each country's currency, we deflate it using the national consumer price index and then convert it into 2011 PPP US dollars (World Bank, 2019b) for cross country comparability. Finally, we include measures of dependency in the household, that is, the number of household members below five and above 65 years of age, who are typically not in the labor force and may require care from other household members.

To evaluate cognitive achievement, we use the standardised percentage of correct answers such that each of these measures have a within country zero mean and unit variance (z-scores). We expect to find a negative correlation between cognitive skills and the likelihood of being NEET as cognitive abilities have been found to be positively correlated with academic and labor market performance (Heckman et al., 2006).

Non cognitive skills are measured by the psychological tests described previously. We follow the literature by using exploratory factor analysis (Krutikova and Lilleør, 2015; Mendolia and Walker, 2015; Schurer, 2017) to capture the latent variation underlying some of these measures. Previous studies have cast doubt on the independence of non cognitive measures like self esteem, locus of control, and self efficacy; that is, that they might be indicators of a common concept (Judge et al., 2002). Following this literature, we construct a core self evaluation index using factor analysis based on these three items as inputs (self esteem, locus of control, and self efficacy). Our empirical model includes the standardised first loading factor estimated using iterated principal factor analysis. This variable reflects the individuals' evaluation of their own abilities and control (Judge et al., 1998). We thus expect to find a negative correlation with the probability of being NEET.

While some socioemotional skills might be influenced by short term events, there are other personality traits that seem to be stable throughout life. This is the case of those measured by the Big Five personality test (Costa and McCrae, 1995; Hampson et al., 2007; Cobb-Clark and Schurer, 2012). The effect of these traits on labor market outcomes such as career success (Gelissen and de Graaf, 2006), earnings (Nyhus and Pons, 2005; Mueller and Plug, 2006), job absenteeism (Judge et al., 1997), and unemployment (Uysal and Pohlmeier, 2011) have been documented extensively in the literature. Our model includes each of the Big Five items — extraversion, agreeableness, conscientiousness, neuroticism and openness — to evaluate whether they are independently correlated with the probability of being NEET, as some of these items have shown to be correlated with factors that are also correlated with being NEET such as academic performance, health habits, the quality of employment relationships, and wages (Furnham et al., 2005; Mueller and Plug, 2006; Hampson et al., 2007). Again, to ease interpretation, each measure was standardised.

_

⁸ In total, there were 30 items, 10 per test. These items show acceptable levels of internal consistency (Cronbach's alpha=0.77).

We also include the standardised results of the Grit test that measures the capability of individuals of achieving long term goals. We expect that individuals scoring high on the test are more likely to be enrolled in education and/or working (Mendolia and Walker, 2014; Light and Nencka, 2019).

To measure educational aspirations, the survey asks individuals about the highest educational degree they would like to complete assuming no constraints. Taking into account the legal structure of each country's educational system, our aspirational measure is the number of additional years of education the respondents would like to complete, beyond what they have completed so far. Having higher educational aspirations is expected to lead to a lower likelihood of beeing NEET (Noh and Lee, 2017).

Expected returns in the labor market also play a role in youths' educational and work decisions (Dominitz and Manski, 1996; Jensen, 2010; Zafar, 2011). To measure the perceived returns to schooling, we include the monthly wage that youths believe tertiary education graduates earn in their local area compared to secondary education graduates (in logs). Reported expected wages are converted to 2011 PPP US dollars after deflating by each country's consumer price index (World Bank, 2019b).

Finally, having imperfect information about educational returns may also affect youths' human capital investment and employment decisions. To measure information biases, we compare expected labor earnings by education level with actual income information obtained from a representative household survey for each country. Actual income was estimated using a comparable population, that is, individuals sorted by education level living in the same city. According to our definition, an individual underestimates the income of any given educational level if the difference between their prediction and the actual income is negative and is equal to at least one standard deviation. The overestimation of labor income is constructed in a similar manner. We use two dummy variables to account for these biases;

[.]

⁹ We used the Pesquisa Nacional por Amostra de Domicílios Contínua 2017 in Brazil; the Encuesta de Caracterización Socioeconómica Nacional 2015 in Chile; the Gran Encuesta Integrada de Hogares 2016 in Colombia; the Encuesta de Hogares de Propósitos Múltiples 2016 in El Salvador; the, Enquête sur les Conditions de Vie des Ménages après Séisme 2012 in Haiti; the Encuesta Nacional de Ocupación y Empleo 2016-2017 in Mexico; and the Encuesta Permanente de Hogares 2016 in Paraguay.

each indicates whether the individual over- or under-estimates the labor earnings associated with the level of education directly above the one the individual currently has.¹⁰

3.3. Descriptive statistics

Table 2 contains the main statistics of the full sample and according to youth's NEET status at the time of the interview. ¹¹ On average, youths in the sample are 19 years old and have completed 11 years of education. The sample is evenly distributed by gender. Thirteen percent had a child or were expecting one before 20 years old. Nearly 18% are neither studying nor working nor engaged in training activities.

NEETs and non NEETs differ in many aspects. NEETs tend to be older and are more likely to be female. They have also completed fewer years of education. Moreover, they belong to poorer households and ones where there are more young children. They are also more likely to have had or been expecting a child during adolescence.¹²

NEETs also show poorer performance in the numeracy and literacy tests and lower scores in the core self evaluation index. Furthermore, NEETs show less extroversion, agreeableness, and openness than other youths, but also less neuroticism. In addition, they have lower educational aspirations and lower wage expectations. Moreover, they are more likely to underestimate the returns to higher levels of education.

In addition, Table 2 shows that NEETs are not idle: they are more likely than non NEETs to engage in domestic chores and in the family business, and to take care of other family members. They are also more likely to be searching for a job. That is, most NEETs, although out of school and out of work, are actively involved in non paid labor and have goals to enter paid labor.

¹⁰ Beliefs about the returns from all levels of education were measured only in Brazil. The Colombian survey lacks information on the beliefs about primary education returns, while the remainder countries have no information on beliefs about graduate studies returns. We use the expected return in the nearest level when the exact information was not available.

¹¹ The descriptive statistics for men and women separately and for each individual country are shown in the Appendix.

¹² Forty percent of female NEETs had a child or were expecting one before 20 years old. In contrast, only 7% of male NEETs were in the same condition. See Tables A1 and A2 in the Appendix.

3.4. Statistical analysis

We estimate a probit model that relates the relevant observable variables with the NEET status as follows:

$$NEET_{i} = \beta_{1} + \beta_{2}X_{i1} + \beta_{3}X_{i2} + \beta_{4}X_{i3} + \beta_{5}X_{i4} + \beta_{6}X_{i5} + \beta_{7}country_{i} + \varepsilon_{i}$$
 (1)

where $NEET_i$ is a dummy that equals one if youth i is a NEET and zero otherwise; X_{i1} is a vector of sociodemographic variables (gender, age, years of education, household's monthly income per capita, teenage parenthood, and number of household members younger than five and older than 65); X_{i2} is a vector of cognitive skills measures (literacy and numeracy); X_{i3} is a vector of non cognitive skill measures (core self evaluation, Grit, and the Big Five personality traits); X_{i4} is the measure of educational aspirations; and X_{i5} is a vector of measures of wage expectations and information biases. We progressively include these sets of variables to better understand any potential correlation among them. In addition, $country_i$ is a set of dummy variables that are equal to one for each country and zero otherwise. We include these dummies to account for cross country differences that affect the work/study decisions of youths, including labor market conditions, the educational environment and social preferences.

We estimate this probit model for the full sample including both men and women, and for each country separately to allow for cross country heterogeneity.¹³

4. Results

4.1. The probability of being a NEET

Marginal effects of probit regressions are reported in Tables 3. Column (1) includes only the sociodemographic variables. Column (2) adds the cognitive skills measures, while column (3) adds the non cognitive skills measures. In turn, column (4) adds the educational

¹³ In the Appendix we report pooled cross country results by gender. See Tables A10 and A11. The results by country and gender are available upon request.

aspirations measure. Finally, column (5) adds the perceived returns to schooling and the information bias dummies.

Being female, years of age, years of education, teenage parenthood, household income, and young children in the household are all significatively associated with the probability of being a NEET (column (1)). For example, women are 5 percentage points (pp) more likely to be NEETs than men. Similarly, the likelihood of being in the NEET status rises 3 pp every year as the individuals age. In turn, each additional year of schooling is related to a 2 pp lower likelihood of being NEET, while having had a child in adolescence is associated with a 12 pp higher probability of currently being a NEET. Interestingly, all estimated marginal effects associated to these socioeconomic variables are practically invariant across models as more variables are included.

As expected, cognitive skills are negatively correlated with the NEET status (column (2)): a unit increase in the z-score of the numeracy and literacy tests are both statistically related to a statistically lower probability of being a NEET (2 pp).

The core self evaluation is also statistically significant and inversely related with the probability of being a NEET youth with a unit increase in the z-score associated with a 2 pp lower probability (column (3)). Among all the Big Five personality traits, only the coefficient on extraversion is always statistically significant and negatively correlated with the dependent variable (a unit increase in the respective z-score is related to a 1 pp lower probability of being a NEET).

Similarly, the educational aspiration variable shows a significant and relevant correlation with NEET status (column (4)). In fact, each additional year of schooling aspired to is related to a 2 pp lower probability of being NEET, a correlation that matches the estimated effect of the actual years of educational achievement.

The remainder of the variables added in column (5) — the return to higher education and the information dummies — show no statistically relevant correlation with the NEET status.

Interestingly, as shown in Tables A10 ad A11 in the Appendix, these correlatory patterns hold when the regression model is separately estimated for men and women.

Table 4 reports the estimated marginal effects of the probit regressions (full specification) in each LAC country. ¹⁴ Overall, large heterogeneity is found in the estimated marginal effects

¹⁴ Small sample sizes might explain why many variables display no statistical significance.

and their significance. For instance, while in Brazil and Paraguay women are 3 pp more likely to be NEETs than men —a non significant effect—, in Mexico they are 11 pp more likely. Similarly, the probability of being in the NEET status shows a larger association with individuals's age in Brazil, El Salvador and Haiti than in the other countries. In addition, in Haiti, years of schooling do not seem to be related to the NEET status while a negative and significant correlation is found in the other countries. Moreover, having had children at an early age in Chile, Mexico or Paraguay is related to a higher probability of being a NEET, however, no significant relationship is found in Brazil, Colombia, El Salvador and Haiti. In turn, once conditioning on all other variables, household income seems to be important in Brazil, Chile, Colombia and Paraguay, but not in El Salvador, Haiti and Mexico. Furthermore, only in Colombia and Mexico there seems to be a correlation with the number of young children in the household.

The marginal effects of cognitive and non cognitive skills on the probability of being a female NEET are heterogeneous across countries as well. Only in Mexico and Paraguay is a unit increase in the z-score of the numeracy test correlated with the NEET status. In turn, literacy skills are statistically significant only in Paraguay. The core self evaluation measure has a statistically significant, marginal negative effect in Chile, Colombia, Mexico and Paraguay. As for the Grit score, a unit increase in the z-score is related to a lower probability of being a NEET in Brazil and El Salvador. The Big Five personality traits are not clearly associated with the NEET condition. In particular, extraversion is important in Chile and Mexico, while neuroticism is in El Salvador and Haiti.

In turn, our educational aspirations variable has a statistically significant and negative correlation with the dependent variable among youths in all countries but El Salvador and Haiti.

Interestingly, we find no statistically significant correlation with expected wages, but we do find a relevant correlation with information biases in Mexico and Paraguay. That is, Mexican youths who overestimate the returns to further education are less likely to be NEETs, while their Paraguayan counterparts who underestimate it are also less likely to be NEETs.

4.2. Discussion

The Millennials in LAC survey shows that NEET young men and women are not merely idle as often portrayed. As a matter of fact, most report allocating time to valuable activities for their households such as helping in domestic tasks and in family businesses. That is, being NEET seems to be much more about certain disadvantages rather than a simple lack of interest in a productive lifestyle.

More specifically, our findings show that, in line with the previous literature, being NEET is correlated with low household income, young dependents in the household, and early parenthood. In addition, this paper shows novel evidence on its correlation with poorer literacy and numeracy skills, a lower evaluation of one's own abilities and control, and lower educational aspirations. These facts raise concerns that NEETs could become disengaged and socially excluded.

This paper also shows that a number of factors that theoretically may affect the time allocation decisions of youths are not correlated with the NEET status in our sample. That is, the Grit test, the agreeableness, conscientiousness, neuroticism, and openness items of the Big Five, and the expected returns to tertiary education are not statistically correlated with being NEET in our estimation results.

Our results have implications for the design of policies aiming to address these concerns. Although we observe cross country heterogeneity in the factors that correlate with the NEET status, all countries may benefit from a number of general policies. For example, programs that alleviate poverty, reduce the prevalence of teen parenthood, provide access to childcare facilities and expand the supply of high quality education and labor market opportunities for youth, all have the potential to reduce educational dropping out and to facilitate the transition into employment (De Hoyos et al., 2016). Since our research also finds impacts from factors other than demographic and socioeconomic conditions, interventions that have proven effective in altering youth's educational and employment trajectories by boosting their socioemotional skills, aspirations, and expectations also have potential to reduce rates of NEET youths. Some of these interventions can be implemented at a low cost.

Non cognitive skills such as self esteem, self efficacy and locus of control are malleable over the life cycle, even more than cognitive skills (Almlund et al., 2011). A number of low cost interventions have shown promising results in affecting these factors such as those aimed

at changing the attitudes of young people towards effort. For instance, Good et al. (2003), Paunesku et al. (2015) and Outes et al. (2017) find a positive impact on the academic performance of randomly selected students who were exposed to scientific evidence proving that the brain is a plastic organ that responds to exercise, and that, therefore, intelligence can be developed through study and practice. Similarly, the evidence in Behncke (2012) and Martins (2010) shows that self efficacy and self esteem can be boosted through words of encouragement, leading to improved student achievement.

The literature provides evidence that aspirations can also be modified at a low cost. More specifically, a set of studies shows that the exposure to individuals who have similar family and social backgrounds and who have succeeded in life can change youths' educational aspirations and their beliefs about their own labor market possibilities. For instance, Nguyen (2010), in Madagscar, randomly assigned role models to students who provided information about his/her family background, educational experience and achievements. Similarly, Dinkelman and Martínez (2014) provided information to Chilean students on how to obtain financing for higher education through messages delivered by mentors of similar socioeconomic status who had already entered tertiary education. Carrell and Sacerdote (2017) randomly assigned college students to high school seniors students to encourage them to pursue higher education in the United States. Finally, Oreopoulos and Petronijevic (2018) randomly assigned mentoring by soon-to-graduate students to younger college students in Canada. All these studies find that fostering youths' knowledge and beliefs that 'people like me can be successful' by exposing them to role models of similar background improve effort and test scores.

In sum, according to our findings, non cognitive skills, aspirations and knowledge about wages, are relevant correlates of young people's educational and employment status even after controlling for various individual and household characteristics. The association of these factors with NEET status are as strong as those of cognitive skills. A promising body of experimental evidence shows that they can be modified through low cost interventions. Hence the economic issue of NEETs — who tend to have very low economic returns over their lifestyle — could be at least partially addressed by said novel low cost interventions along with the more traditional but more expensive ones of addressing teenage parenthood, dropout rates, and the like.

Our work has a number of shortcomings. First due to the cross sectional nature of our dataset, we cannot establish any causal relationship between the variables of interest and the probability of being NEET. In fact, most of the variables included in the analysis might potentially be either determinants or effects of the NEET condition. Future analyses should attempt to disentangle the causal effect of skills, aspirations and expectations on the likelihood of being NEET.

Second, the decision or outcome of being neither working nor studying is most likely a dynamic one. Our dataset allows us to capture the role of past outcomes, such as previous investments in human capital, and of future expectations. However, we are not able to analyze how these evolve over time jointly with education and employment decisions. Future work should rely on panel data sets to better understand these dynamics.¹⁵ In addition, the availability of larger sample sizes might help obtain more precise estimates and understand whether some of our results are due to small sample sizes.

Finally, our paper has analysed the association of the NEET status with a limited set of measures of cognitive and non cognitive skills, and aspirations. However, the literature offers a much larger set of measures of cognitive abilities, personality traits and motivation (Kautz et al., 2014). Future work should further analyse which measures are best correlated with being out of work and study.

5. Conclusion

In this paper we take advantage of a novel dataset in order to examine the factors that correlate with young people who are neither involved in education, training, nor the labor market (NEETs). This rich dataset that covers a large sample of Latin American and Caribbean young men and women, allows us to focus on the role of cognitive and non cognitive skills, and of expectations and aspirations. The relevance of these factors in the context of developing countries remains unclear as most of the literature has emphasised the role of socioeconomic conditions in the educational and labor market context of richer countries.

- _

¹⁵ Favara and Sánchez (2018) obtain similar results using longitudinal data from Peru.

Our paper thus is an attempt to fill this gap. We identify a number of additional factors that correlate with the NEET status in LAC. In our analysis, cognitive skills, core self evaluation, extraversion and educational aspirations are the most common correlates. We also discuss how to use these findings to inform the design of policies to address this problem in Latin America and the Caribbean.

In spite of these limitations, the richness of the Millennials in LAC survey has allowed us to offer a thorough description of this population in this paper and to identify factors that correlate with the NEET status. We believe that our findings are useful for the design of educational and labor market policies aimed at young people's social inclusion.

Acknowledgements

We would like to thank all the individuals and institutions who contributed to the project "Millennials in Latin America and the Caribbean: to study or work?", from which we obtained the data used in this paper. We also want to thank an anonymous referee for valuable feedback. The authors are responsible for all results and views, which do not necessarily represent the view of any of the institutions where they are affiliated.

Data Statement

The data that support the findings of this study are available from the corresponding author upon reasonable request.

References

Alcázar, L., Balarin, M., Glave, C., & Rodríguez, M. F. (2019). Fractured lives: understanding urban youth vulnerability in Perú. *Journal of Youth Studies*. 10.1080/13676261.2019.1587154.

Alfieri, S., Sironi, E., Marta, E., Rosina, A., & Marzana, D. (2015). Young Italian NEETs (not in employment, education, or training) and the influence of their family background. *Europe's Journal of Psychology*. 10.5964/ejop.v11i2.901.

Almlund, M., Duckworth, A.L., Heckman, J.J., & Kautz, T. (2011). *Personality Psychology and Economics*. In E. Hanushek, S. Machin, and L. Woessman, eds., Handbook of the Economics of Education, Amsterdam: Elsevier. pp. 1-181.

- Attanasio, O. P., & Kaufmann, K. M. (2014). Education choices and returns to schooling: Mothers' and youths' subjective expectations and their role by gender. *Journal of Development Economics*, 109, 203-216.
- Azevedo, J. P., Favara, M., Haddock, S. E., Lopez-Calva, L. F., Muller, M., & Perova, E. (2013). *Teenage pregnancy and opportunities in Latin America and the Caribbean on early child bearing, poverty, and economic achievement: Embarazo adolescente resumen 2013 (Spanish)*. Washington, DC: World Bank Group.
- Baggio, S., Iglesias, K., Deline, S., Studer, J., Henchoz, Y., Mohler-Kuo, M., & Gmel, G. (2015). Not in education, employment, or training status among young swiss men. Longitudinal associations with mental health and substance Use. *Journal of Adolescent Health*. 10.1016/j.jadohealth.2014.09.006.
- Baland, J. M., & Robinson, J. A. (2000). Is child labor inefficient? *Journal of Political Economy*, 108(4), 663-679.
- Baron, J. D., Popova, A., & Sánchez, A. (2016). *Following Mexican youth: A short-run study of time use decisions*. Policy Research Working Paper, No. 7534. World Bank, Washington, DC. World Bank.
- Basu, K., & Van, P. H. (1998). The economics of child labor. *American Economic Review*, 88(3) 412-427.
- Behncke, S. (2012). How do shocks to con-cognitive skills affect test scores? *Annals of Economics and Statistics*, 107/108, 155-173.
- Behrman, J., de Hoyos, R., & Székely, M. (2015). Out of school and out of work: A conceptual framework for investigating "ninis" in Latin America and the Caribbean. Washington, DC: World Bank. Background paper for the "Out of School, Out of Work" study. https://openknowledge.worldbank.org/handle/10986/22349.
- Bernal, R., Pulido, X., Sánchez, F., & Sánchez, L.M. (2018). Decisiones de vida de los jóvenes en Bogotá: ¿pobreza, habilidades o comportamientos riesgosos? In Novella, R., Repetto, A., Robino, C. and Rucci, G. (eds.) *Millennials en América Latina y el Caribe: ¿Trabajar o estudiar?* Washington, DC: IADB.
- Carrell, S., & Sacerdote, B. (2017). Why do college-going interventions work? *American Economic Journal: Applied Economics*, 9(3), 124–151.
- Cobb-Clark, D.A., & Schurer, S. (2012). The stability of big-five personality traits. Economics Letters, 115(1), 11-15. ISSN 0165-1765, https://doi.org/10.1016/j.econlet.2011.11.015.
- Cornaglia, F., Crivellaro, E., & McNally,S. (2015). Mental health and education decisions. *Labour Economics*, *33*, 1-12. ISSN 0927-5371. https://doi.org/10.1016/j.labeco.2015.01.005.

Costa, P., & McCrae, R. (1995). Solid ground in the wetlands of personality: A reply to Block. *Psychological Bulletin*, 117, 216-20; discussion 226. 10.1037//0033-2909.117.2.216.

Crawford, C., Duckworth, K., Vignoles, A., Wyness, G. (2011) *Young People's Education and Labour Market Choices Aged 16/17 to 18/19*. Department of Education. Research Report DFE-RR182.

De Hoyos, R., Rogers, H., & Székely, M. (2016). Out of school and out of work: Risk and opportunities for Latin America's ninis. World Bank, Washington, DC.

Dinkelman, T., & Martinez, C.A. (2014). Investing in schooling in Chile: The role of information about financial aid for higher education. *Review of Economics and Statistics*, 96(2), 244-257.

Dominitz, J., & Manski, C.F. (1996). Eliciting student expectations of the returns to schooling. *Journal of Human Resources*, 31(1), 1–26.

Duckworth, A. L., & Quinn, P. D. (2009). Development and validation of the Short GRIT Scale (Grit-S). *Journal of Personality Assessment*, 91(2), 166-174.

Edmonds, E. (2008). Child labor. *Handbook of Development Economics*, 4, 3607-3709, Elsevier.

Favara, M., & Sánchez, A. (2018). La transición hacia el mercado laboral y los etsudios postsecundarios en Perú: Evidencia del estudio Niños del Milenio. In Novella, R., Repetto, A., Robino, C. and Rucci, G. (eds.) *Millennials en América Latina y el Caribe: ¿Trabajar o estudiar?* Washington, DC: IADB.

Furnham, A., Petrides, K. V., Tsaousis, I., Konstantinos, P., & Garrod, D. (2005). A cross-cultural investigation into the relationships between personality traits and work values. *The Journal of Psychology*, *139*, 5-32. 10.3200/JRLP.139.1.5-32.

Gelissen, J., & de Graaf, P. M. (2006). Personality, social background, and occupational career success. *Social Science Research*, *35*(3), 702-726, ISSN 0049-089X.

Goldman-Mellor, S., Caspi, A., Arseneault, L., Ajala, N., Ambler, A., Danese, A., Fisher, H., Hucker, A., Odgers, C., Williams, T., Wong, C., & Moffitt, T. E. (2016). Committed to work but vulnerable: self perceptions and mental health in NEET 18-year olds from a contemporary British cohort. *Journal of Child Psychology and Psychiatry, and Allied Disciplines*, 57(2), 196-203.

Good, C., Aronson, J., & Inzlicht, M. (2003). Improving adolescents' standardised test performance: An intervention to reduce the effects of stereotype threat. *Applied Developmental Psychology*, 24, 645–662.

Hampson, S. E., Goldberg, L. R., Vogt, T. M., & Dubanoski, J. P. (2007). Mechanisms by which childhood personality traits influence adult health status: Educational attainment and healthy behaviors. *Health Psychology*, 26(1), 121-125.

Heckman, J. J., Stixrud, J., & Urzúa, S. (2006). The effects of cognitive and noncognitive abilities on labor market. *Journal of Labor Economics*, 24(3), 411-482.

International Labour Organization (ILO). (2015). What does NEETs mean and why is the concept so easily misinterpreted? Technical Brief No.1.

Jensen, R. (2010). The (perceived) returns to education and the demand for schooling. *The Quarterly Journal of Economics*, 125, 515–548.

Jongbloed, J & Giret, J. F. (2019). The variables roles of skill and education in predicting youth NEET statuses: Gender differences in comparative perspective. Mimeo.

Judge, T. A., Martocchio, J., & Thoresen, C. J. (1997). Five-factor model of personality and employee absence. *Journal of Applied Psychology*, 82, 745-755. 10.1037//0021-9010.82.5.745.

Judge, T. A., Locke, E. A., Durham, C.C., & Kluger, A. N. (1998). Dispositional effects on job and life satisfaction: The role of core evaluations. *The Journal of Applied Psychology*, 83, 17-34. 10.1037/0021-9010.83.1.17.

Judge, T. A., Erez, A., Bono, J. E., & Thoresen, C. J. (2002). Are measures of self esteem, neuroticism, locus of control, and generalized self efficacy indicators of a common core construct? *Journal of Personality and Social Psychology*, 83(3), 693-710.

Kautz, T., Heckman, J. J., Diris, R., Ter Weel, B., & Borghans, L. (2014). Fostering and measuring skills: Improving cognitive and non cognitive skills to promote lifetime success (No. w20749). National Bureau of Economic Research.

Krumpal, I. (2013). Determinants of social desirability bias in sensitive surveys: A literature review. *Quality and Quantity*, 47, 2025–2047.

Krutikova, S., & Lilleør, H. (2015). Fetal origins of personality: Effects of early life circumstances on adult personality traits. CSAE Working Paper Series 2015-03, Centre for the Study of African Economies, University of Oxford.

Light, A., & Nencka, P. (2019). Predicting educational attainment: Does grit compensate for low levels of cognitive ability? *Learning and Individual Differences*, 70, 142-154. ISSN 1041-6080.

Martins, P.S. (2010). *Can targeted, non cognitive skills programs improve achievement?* IZA Discussion Paper No. 5266.

Mascherini, M., Salvatore, L., Meierkord, A., & Jungblut, J. M. (2012). *NEETs: Young people not in employment, education or training: Characteristics, costs and policy responses in Europe*. Luxembourg: Publications Office of the European Union.

Mendolia, S., & Walker, I. (2014). *Do NEETs need grit?* IZA Discussion Papers 8740, Institute for the Study of Labor (IZA).

Mendolia, S., & Walker, I. (2015). Youth unemployment and personality traits. *IZA Journal of Labor Economics*, 4(19), 1-26.

Mueller, G., & Plug, E. (2006). Estimating the effect of personality on male and female earnings. *Industrial and Labor Relations Review*, 60(1), 3-22.

Ng-Knight, T., & Schoon I. (2017). Can locus of control compensate for socioeconomic adversity in the transition from school to work? *Journal of Youth Adolescence*, 46, 2114–28.

Nguyen, T. (2010). *Information, role models and perceived returns to education: Experimental evidence from Madagascar.* Abdul Latif Jameel Poverty Action Lab (JPAL).

Noh, H., & Lee, B. J. (2017). Risk factors of NEET (not in employment, education or training) in South Korea: an empirical study using panel data. *Asia Pacific Journal of Social Work and Development.* 27, 28-38. 10.1080/02185385.2017.1289860.

Novella, R., Repetto, A., Robino, C., & Rucci, G. (2018). *Millennials en América Latina y el Caribe: ¡Trabajar o estudiar?* Washington, DC: IADB.

Novella, R., & Repetto, A. (2019). Accompanying youth in education and work: what works and what does not? A review of evaluations of low cost interventions. Working Paper, Universidad Adolfo Ibáñez.

Nyhus, E. K., & Pons, E. (2005). The effects of personality on earnings. *Journal of Economic Psychology*, 26, 3, 363-384. ISSN 0167-4870.

OECD (2014). OECD Employment Outlook 2014. Paris: OECD Publishing, 2011.

OECD (2015a). *Skills for Social Progress: The Power of Social, and Emotional Skills*. Paris: OECD Publishing, 2011.

OECD (2015b). OECD Skills Outlook 2015: Youth, Skills and Employability. Paris: OECD Publishing, 2011.

Oreopoulos, P., & Petronijevic, U. (2018). Student coaching: How far can technology go? *Journal of Human Resources*, 53(2), 299-329.

Outes, I., Sánchez, A., & Vakis, R. (2017). Cambiando la mentalidad de los estudiantes: evaluación de impacto de Expande tu Mente sobre el rendimiento académico en tres regiones del Perú. Documento de Investigación 83, GRADE.

Pallais, A. (2014). Inefficient hiring in entry-level labor markets. *American Economic Review*, 104(11), 3565-99.

Paunesku, D., Walton, G.M., Romero, C., Smith, E.N., Yeager, D.S., & Dweck C.S. (2015). Mind-set interventions are a scalable treatment for academic underachievement. *Psychological Science*, *26*, 784 –793.

Rammstedt, B., & P. John, O. (2007). Measuring personality in one minute or less: A 10-item short version of the Big Five Inventory in English and German. *Journal of Research in Personality*, 41, 203-212. 10.1016/j.jrp.2006.02.001.

Rosenberg, M. (1965). *Society and the adolescent self image*. Princeton, NJ: Princeton University Press.

Rotter, J. (1966). Generalized expectancies for internal versus external control of reinforcement. *Psychological Monographs: General and Applied*, 80(1), 1-28.

Salvà-Mut, F., Tugores-Ques, M., & Quintana-Murci, E. (2018). NEETs in Spain: an analysis in a context of economic crisis. *International Journal of Lifelong Education*, *37*(2), 168-183.

Schwarzer, R., & Jerusalem, M. (1995). *Generalized self efficacy scale*. In J. Weinman, S. Wright, and M. Johnston, *Measures in health psychology: A user's portfolio*. Causal and control beliefs (pp. 35-37). Windsor, UK: NFER-NELSON.

Schultz, T. W. (1960). Capital formation by education. *Journal of Political Economy*, 68(6), 571-583.

Schurer, S. (2017). Bouncing back from health shocks: Locus of control and labor supply. *Journal of Economic Behavior and Organization*, *133*, 1-20, ISSN 0167-2681.

Skoufias, E. & Parker, S. (2001). Conditional cash transfers and their impact on child work and schooling: Evidence from the PROGRESA program in Mexico. *Economía*, 2, 45-86.

STATA StataCorp (2017). Stata Statistical Software: Release 15. College Station, TX: StataCorp LP, USA.

Susanli, B.Z. (2016). Understanding the neet in Turkey. *Eurasian Journal of Economics and Finance*, 4, 42-57. 10.15604/ejef.2016.04.02.004.

Székely, M., & Karver, J. G. (2015). *Youth out of school and out of work in Latin America: a cohort approach (English)*. Policy Research working paper; no. WPS 7421. Washington, D.C.: World Bank Group.

Tourangeau, R., Rasinski, K., Jobe, J.B., Smith, T.W., & Pratt, W.F. (1997). Sources of error in a survey on sexual behavior. *Journal of Official Statistics*, 13(4), 341-365.

Uysal, S. D., & Pohlmeier, W. (2011). Unemployment duration and personality. *Journal of Economic Psychology*, 32(6), 980-992. ISSN 0167-4870.

Wickremeratne, N., & Dunusinghe, P. (2018). Youth not in education, employment and training (NEET) in Sri Lanka. Advances in Economics and Business, 6, 339-352.

Wiswall, M., & Zafar, B. (2015). Determinants of college major choice: Identification using an information experiment. *The Review of Economic Studies*, 82(2), 791-824.

World Bank. (2019a). Share of youth not in education, employment or training, total (% of youth population) [Data file]. Retrieved from https://data.worldbank.org/indicator/SL.UEM.NEET.ZS

World Bank. (2019b). *International Comparison Program database*. [Data file]. Retrieved from https://data.worldbank.org/indicator/PA.NUS.PPP

Yates, S., Harris, A., Sabates, R., & Staff, J. (2011). Early occupational aspirations and fractured transitions: A study of entry into 'NEET' status in the UK. *Journal of Social Policy*, 40(3), 513-534. doi:10.1017/S0047279410000656

Zafar, B. (2011). How do college students form expectations? *Journal of Labor Economics*, 29(2), 311–348.

Zudina, A. (2017). What makes youth become NEET? The evidence from Russian LFS. Higher School of Economics Research Paper No. WP BRP 177/EC/2017. Insert Figure 1 here

Table 1. Millennials in LAC Survey

Country	Sample size (youths)	Area	Data collection period
Brazil	1488	Urban areas of Recife	April-May, 2018
Chile	3560	Urban areas of Santiago, Valparaíso, and Biobío	July-October, 2017
Colombia	1500	Urban areas of Bogotá	May-August, 2017
El	Metropolitan area of San		February-March, 2017
Salvador	1772	Salvador	reordary waren, 2017
Haiti	860	Metropolitan area of Port-au- Prince, Carrefour, and Pétion- ville	April-May, 2018
Mexico	2064 ^a	Mexico City and its metropolitan area	May-September, 2017
Paraguay	1536	Asunción and urban areas of eight Central Department districts	May-June, 2017

^a In addition, the survey in Mexico included 1,320 responsible adults in the household (the father, mother, or other adult in charge of the youth's care if he/she is 17 or younger).

Table 2. Descriptive statistics

	A	11	NE	ET	Not NEET		Sig.	
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Means test d	
Age	19.3	(0.03)	20.5	(0.06)	19.0	(0.03)	***	
Male ^a	0.47	(0.01)	0.33	(0.01)	0.50	(0.01)	***	
Years of education	11.10	(0.03)	10.8	(0.07)	11.2	(0.03)	***	
Teenage parenthood ^a	0.13	(0.00)	0.29	(0.01)	0.09	(0.00)	***	
Household monthly income per capita (log) ^b Number of household members younger than 5	238.9	(2.63)	184.9	(5.13)	250.4	(2.98)	***	
years old	0.31	(0.01)	0.54	(0.02)	0.26	(0.01)	***	
Number of household members older than 65 years old	0.20	(0.01)	0.17	(0.01)	0.20	(0.01)	**	
Numeracy skills (z-score) ^c	0.03	(0.01)	-0.17	(0.03)	0.07	(0.01)	***	
Literacy skills (z-score) ^c	0.05	(0.01)	-0.14	(0.03)	0.09	(0.01)	***	
Core self evaluation (z-score) ^c	0.04	(0.01)	-0.15	(0.03)	0.08	(0.01)	***	
Grit test (z-score) ^c	0.04	(0.01)	-0.07	(0.03)	0.06	(0.01)	***	
Big Five								
Extraversion (z-score) ^c	0.01	(0.01)	-0.10	(0.03)	0.03	(0.01)	***	
Agreeableness (z-score) ^c	0.00	(0.01)	-0.06	(0.03)	0.02	(0.01)	**	
Conscientiousness (z-score) ^c	0.02	(0.01)	0.03	(0.02)	0.02	(0.01)		
Neuroticism (z-score) ^c	0.02	(0.01)	-0.03	(0.03)	0.03	(0.01)	*	
Openness (z-score) ^c	0.02	(0.01)	-0.05	(0.03)	0.04	(0.01)	***	
Education aspirations (additional years)	5.43	(0.03)	4.84	(0.07)	5.55	(0.03)	***	
Expected return to tertiary education (log)	0.94	(0.01)	0.91	(0.02)	0.94	(0.01)	**	
Overestimates salary ^a	0.06	(0.00)	0.04	(0.01)	0.06	(0.00)	***	
Underestimates salary ^a	0.09	(0.00)	0.11	(0.01)	0.09	(0.00)	***	
Activities								
Looking for a job ^a	0.15	(0.00)	0.39	(0.01)	0.09	(0.00)	***	
Family caregiver ^a	0.42	(0.01)	0.58	(0.01)	0.38	(0.01)	***	
Domestic work/family business ^a	0.84	(0.00)	0.89	(0.01)	0.83	(0.00)	***	
Person with disability ^a	0.02	(0.00)	0.02	(0.00)	0.02	(0.00)		
Number of observations	82	76	14	60	68	16		

^a The fraction of the corresponding group is shown.

^b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%, ***significant at 1%.

Table 3. Marginal effects of probit regressions: probability of being NEET

	(1)	(2)	(3)	(4)	(5)
Men=1	-0.05***	-0.05***	-0.05***	-0.06***	-0.06***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Age	0.03***	0.03***	0.03***	0.03***	0.03***
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Years of education	-0.02***	-0.01***	-0.01***	-0.02***	-0.02***
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Teenage parenthood=1	0.12***	0.12***	0.12***	0.11***	0.11***
	(0.02)	(0.02)	(0.02)	(0.01)	(0.01)
Household monthly income per capita (log) ^a	-0.04***	-0.03***	-0.03***	-0.03***	-0.03***
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Number of household members younger than 5	0.02***	0.02***	0.02***	0.02***	0.02***
, ,	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Number of household members older than 65	-0.01	-0.01	-0.01	-0.00	-0.00
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Numeracy skills (z-score) ^b		-0.02***	-0.01***	-0.01***	-0.01***
		(0.00)	(0.00)	(0.00)	(0.00)
Literacy skills (z-score) ^b		-0.02***	-0.01***	-0.01***	-0.01***
		(0.00)	(0.00)	(0.00)	(0.00)
Core self evaluation (z-score) ^b			-0.02***	-0.02***	-0.02***
			(0.00)	(0.00)	(0.00)
Grit test (z-score) ^b			-0.01	-0.01	-0.01
			(0.00)	(0.00)	(0.00)
Big Five					
Extraversion (z-score) ^b			-0.01***	-0.01***	-0.01***
			(0.00)	(0.00)	(0.00)
Agreeableness (z-score) ^b			-0.00	-0.00	-0.00
			(0.00)	(0.00)	(0.00)
Conscientiousness (z-score) ^b			-0.00	-0.00	-0.00
` ,			(0.00)	(0.00)	(0.00)
Neuroticism (z-score) ^b			0.01	0.01	0.01
,			(0.00)	(0.00)	(0.00)
Openness (z-score) ^b			-0.00	-0.00	-0.00
1 /			(0.00)	(0.00)	(0.00)
Education aspirations (additional years)				-0.02***	-0.02***
				(0.00)	(0.00)
Expected return to tertiary education (log)					-0.00
					(0.01)
Overestimates salary=1					-0.02
•					(0.02)
Underestimates salary=1					0.01

					(0.01)
Country dummies (base=Brazil)					
Chile	-0.04**	-0.05**	-0.05**	-0.06***	-0.06***
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Colombia	-0.02	-0.03	-0.03	-0.04*	-0.03
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
El Salvador	0.01	0.00	0.01	-0.00	-0.00
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Haiti	-0.10***	-0.09***	-0.09***	-0.10***	-0.10***
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Mexico	-0.00	-0.00	-0.00	-0.01	-0.01
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Paraguay	-0.00	-0.01	-0.01	-0.01	-0.01
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Observations	8276	8276	8276	8276	8276
Pseudo R ²	0.12	0.12	0.13	0.14	0.14

Source: Millennials in LAC survey.

Note: Standard errors in parentheses. *Significant at 10%, ** significant at 5%, ***significant at 1%.

^a Household monthly income per capita and expected wages were deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^b Standardized measure.

Table 4. Marginal effects of probit regressions: probability of being NEET by country

	Brazil	Chile	Colombia	El Salvador	Haiti	Mexico	Paraguay
Men=1	-0.03	-0.04***	-0.06***	-0.05*	-0.06*	-0.11***	-0.03
	(0.04)	(0.01)	(0.02)	(0.03)	(0.03)	(0.02)	(0.02)
Age	0.04***	0.02***	0.02***	0.04***	0.04***	0.02***	0.02***
	(0.01)	(0.00)	(0.00)	(0.01)	(0.01)	(0.00)	(0.00)
Years of education	-0.03***	-0.01***	-0.03***	-0.03***	-0.01	-0.03***	-0.01*
	(0.01)	(0.00)	(0.01)	(0.01)	(0.01)	(0.00)	(0.01)
Teenage parenthood=1	0.06	0.12***	0.03	-0.01	0.17	0.13***	0.15***
	(0.07)	(0.03)	(0.03)	(0.05)	(0.11)	(0.03)	(0.04)
Household monthly income per capita (log) ^a	-0.04*	-0.04***	-0.04***	-0.03	-0.02	-0.02	-0.03*
	(0.02)	(0.01)	(0.01)	(0.02)	(0.01)	(0.01)	(0.02)
Number of household members younger than 5	0.04	0.00	0.04**	-0.00	0.00	0.05***	0.02
	(0.03)	(0.01)	(0.02)	(0.03)	(0.03)	(0.02)	(0.02)
Number of household members older than 65	0.03	-0.02	-0.02	0.03	-0.02	0.00	-0.00
	(0.04)	(0.01)	(0.02)	(0.03)	(0.02)	(0.03)	(0.02)
Numeracy skills (z-score) ^b	-0.01	-0.00	-0.01	-0.01	-0.02	-0.02**	-0.02*
, , , ,	(0.02)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.01)
Literacy skills (z-score) ^b	-0.00	-0.01	-0.01	-0.03	-0.01	-0.00	-0.02**
, , ,	(0.02)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.01)
Core self evaluation (z-score) ^b	0.02	-0.02***	-0.03**	-0.02	-0.00	-0.02**	-0.02*
,	(0.02)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.01)
Grit test (z-score) ^b	-0.05**	-0.00	0.00	-0.03*	0.01	-0.00	-0.01
,	(0.02)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.01)
Big Five							
Extraversion (z-score) ^b	0.00	-0.01**	-0.01	-0.02	-0.01	-0.02*	0.01
,	(0.02)	(0.01)	(0.01)	(0.01)	(0.02)	(0.01)	(0.01)
Agreeableness (z-score) ^b	0.01	0.00	-0.00	-0.01	-0.02	-0.00	0.01
,	(0.02)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.01)
Conscientiousness (z-score) ^b	-0.02	-0.01	-0.00	0.02	0.03	-0.01	-0.01
` ,	(0.02)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.01)
Neuroticism (z-score) ^b	-0.03	0.01	0.01	0.03*	-0.05***	0.00	0.01
,	(0.02)	(0.01)	(0.01)	(0.02)	(0.02)	(0.01)	(0.01)
Openness (z-score) ^b	0.01	-0.00	0.00	0.02	-0.02	-0.00	-0.01
,	(0.02)	(0.01)	(0.01)	(0.01)	(0.02)	(0.01)	(0.01)
Education aspirations (additional years)	-0.02***	-0.02***	-0.02***	-0.01	-0.00	-0.01**	-0.01**
,	(0.01)	(0.00)	(0.01)	(0.01)	(0.01)	(0.00)	(0.01)
Expected return to tertiary education (log)	0.01	-0.01	-0.00	0.01	0.00	-0.01	0.01
	(0.04)	(0.01)	(0.02)	(0.04)	(0.02)	(0.02)	(0.02)
Overestimates salary=1	0.13	0.02	-0.01	0.00	0.00	-0.08**	-0.08

	(0.19)	(0.03)	(0.03)	(0.09)	(0.09)	(0.04)	(0.06)
Underestimates salary=1	-0.09	-0.07	-0.07	0.05	-0.02	0.03	-0.07**
·	(0.08)	(0.05)	(0.05)	(0.07)	(0.10)	(0.02)	(0.04)
Observations	428	2738	1314	687	479	1498	1132
Pseudo R ²	0.17	0.15	0.14	0.13	0.19	0.20	0.12

Note: Standard errors in parentheses. *Significant at 10%, ** significant at 5%, ***significant at 1%.

a Household monthly income per capita and expected wages were deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^b Standardized measure.

Table A1. Descriptive statistics – Women

	A	.11	NE	ET	No N	Sig.	
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Mean test d
Age	19.5	(0.04)	20.8	(0.08)	19.1	(0.05)	***
Male ^a	0.00	(0.00)	0.00	(0.00)	0.00	(0.00)	
Years of education	11.1	(0.04)	10.8	(0.08)	11.3	(0.04)	***
Teenage parenthood ^a	0.19	(0.01)	0.40	(0.02)	0.13	(0.01)	***
Household monthly income per capita (log) ^b	224.0	(3.47)	179.0	(6.54)	237.2	(4.03)	***
Number of household members younger than 5 years old	0.41	(0.01)	0.71	(0.03)	0.33	(0.01)	***
Number of household members older than 65 years old	0.18	(0.01)	0.15	(0.01)	0.19	(0.01)	**
Numeracy skills (z-score) ^c	-0.07	(0.02)	-0.20	(0.03)	-0.03	(0.02)	***
Literacy skills (z-score) ^c	0.05	(0.01)	-0.11	(0.03)	0.10	(0.02)	***
Core self evaluation (z-score) ^c	-0.01	(0.01)	-0.16	(0.03)	0.03	(0.02)	***
Grit test (z-score) ^c	0.06	(0.02)	-0.03	(0.03)	0.08	(0.02)	***
Big Five							
Extraversion (z-score) ^c	-0.04	(0.02)	-0.14	(0.03)	-0.01	(0.02)	***
Agreeableness (z-score) ^c	0.01	(0.02)	-0.06	(0.03)	0.04	(0.02)	***
Conscientiousness (z-score) ^c	0.07	(0.01)	0.07	(0.03)	0.07	(0.02)	
Neuroticism (z-score) ^c	-0.16	(0.02)	-0.11	(0.03)	-0.17	(0.02)	
Openness (z-score) ^c	0.05	(0.01)	-0.03	(0.03)	0.07	(0.02)	***
Education aspirations (additional years)	5.42	(0.04)	4.90	(0.09)	5.58	(0.05)	***
Expected return to tertiary education (log)	0.93	(0.01)	0.90	(0.02)	0.94	(0.01)	**
Overestimates salary ^a	0.05	(0.00)	0.03	(0.01)	0.05	(0.00)	***
Underestimates salary ^a	0.11	(0.00)	0.13	(0.01)	0.10	(0.01)	**
Activities							
Looking for a job ^a	0.16	(0.01)	0.35	(0.02)	0.10	(0.01)	***
Family caregiver ^a	0.51	(0.01)	0.70	(0.01)	0.45	(0.01)	***
Domestic work/family business ^a	0.88	(0.00)	0.94	(0.01)	0.86	(0.01)	***
Person with disability ^a	0.02	(0.00)	0.02	(0.01)	0.02	(0.00)	
Number of observations	43	662	98	33	33	79	

^a The fraction of the corresponding group is shown.

^b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%,

^{***}significant at 1%.

Table A2. Descriptive statistics – Men

	I	A 11	NE	ET	No N	Sig.	
Variables	Mea n	(SE)	Mean	(SE)	Mean	(SE)	Mean test ^d
Age	19.1	(0.04)	19.9	(0.11)	19.0	(0.05)	***
Male ^a	1.00	(0.00)	1.00	(0.00)	1.00	(0.00)	
Years of education	11.0	(0.04)	10.9	(0.11)	11.1	(0.04)	*
Teenage parenthood ^a	0.05	(0.00)	0.07	(0.01)	0.05	(0.00)	*
Household monthly income per capita (log) ^b Number of household members younger than 5	255.4	(3.98)	197.1	(8.03)	263.5	(4.37)	***
years old	0.19	(0.01)	0.20	(0.02)	0.19	(0.01)	
Number of household members older than 65 years old	0.22	(0.01)	0.22	(0.02)	0.22	(0.01)	
Numeracy skills (z-score) ^c	0.13	(0.02)	-0.11	(0.05)	0.17	(0.02)	***
Literacy skills (z-score) ^c	0.05	(0.02)	-0.21	(0.05)	0.08	(0.02)	***
Core self evaluation (z-score) ^c	0.09	(0.02)	-0.15	(0.05)	0.12	(0.02)	***
Grit test (z-score) ^c	0.01	(0.02)	-0.14	(0.04)	0.04	(0.02)	***
Big Five							
Extraversion (z-score) ^c	0.06	(0.02)	-0.02	(0.04)	0.07	(0.02)	*
Agreeableness (z-score) ^c	-0.01	(0.02)	-0.05	(0.05)	-0.01	(0.02)	
Conscientiousness (z-score) ^c	-0.03	(0.02)	-0.06	(0.04)	-0.03	(0.02)	
Neuroticism (z-score) ^c	0.21	(0.02)	0.16	(0.04)	0.22	(0.02)	
Openness (z-score) ^c	0.00	(0.02)	-0.09	(0.04)	0.01	(0.02)	*
Education aspirations (additional years)	5.43	(0.04)	4.71	(0.12)	5.53	(0.05)	***
Expected return to tertiary education (log)	0.94	(0.01)	0.92	(0.03)	0.95	(0.01)	
Overestimates salary ^a	0.07	(0.00)	0.07	(0.01)	0.07	(0.00)	
Underestimates salary ^a	0.08	(0.00)	0.08	(0.01)	0.08	(0.00)	
Activities							
Looking for a job ^a	0.13	(0.01)	0.48	(0.02)	0.08	(0.00)	***
Family caregiver ^a	0.31	(0.01)	0.33	(0.02)	0.31	(0.01)	
Domestic work/family business ^a	0.79	(0.01)	0.80	(0.02)	0.79	(0.01)	
Person with disability ^a	0.02	(0.00)	0.02	(0.01)	0.03	(0.00)	
Number of observations	39	914	4	77	34	37	

^a The fraction of the corresponding group is shown.

^b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%, ***significant at 1%.

Table A3. Descriptive statistics – Brazil

	A	.11	NI	EET	No NEET		Sig.
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Mean test d
Age	19.4	(0.14)	20.9	(0.23)	19.0	(0.16)	***
Malea	0.53	(0.02)	0.47	(0.05)	0.55	(0.03)	
Years of education	10.4	(0.13)	10.1	(0.29)	10.5	(0.14)	
Teenage parenthood ^a	0.11	(0.01)	0.22	(0.04)	0.07	(0.01)	***
Household monthly income per capita (log) ^b	211.6	(9.97)	178.2	(19.87)	220.6	(11.43)	*
Number of household members younger than 5 years old	0.23	(0.03)	0.40	(0.08)	0.19	(0.02)	**
Number of household members older than 65 years old	0.21	(0.02)	0.22	(0.05)	0.20	(0.03)	
Numeracy skills (z-score) ^c	0.05	(0.05)	-0.21	(0.11)	0.12	(0.06)	***
Literacy skills (z-score) ^c	0.04	(0.04)	-0.07	(0.10)	0.07	(0.05)	
Core self evaluation (z-score) ^c	0.00	(0.05)	-0.14	(0.10)	0.04	(0.05)	
Grit test (z-score) ^c	0.00	(0.05)	-0.24	(0.09)	0.06	(0.06)	***
Big Five							
Extraversion (z-score) ^c	-0.05	(0.05)	-0.08	(0.10)	-0.04	(0.05)	
Agreeableness (z-score) ^c	-0.01	(0.05)	-0.08	(0.10)	0.00	(0.06)	
Conscientiousness (z-score) ^c	-0.04	(0.05)	-0.18	(0.09)	0.00	(0.06)	*
Neuroticism (z-score) ^c	-0.07	(0.05)	-0.08	(0.09)	-0.07	(0.06)	
Openness (z-score) ^c	-0.03	(0.05)	-0.11	(0.10)	-0.01	(0.06)	
Education aspirations (additional years)	6.38	(0.15)	5.31	(0.36)	6.67	(0.16)	***
Expected return to tertiary education (log)	0.72	(0.02)	0.64	(0.04)	0.74	(0.02)	*
Overestimates salary ^a	0.01	(0.01)	0.02	(0.02)	0.01	(0.01)	
Underestimates salary ^a	0.02	(0.01)	0.01	(0.01)	0.02	(0.01)	
Activities							
Looking for a job ^a	0.23	(0.02)	0.44	(0.05)	0.18	(0.02)	***
Family caregiver ^a	0.29	(0.02)	0.40	(0.05)	0.26	(0.02)	**
Domestic work/family business ^a	0.75	(0.02)	0.72	(0.05)	0.76	(0.02)	
Person with disability ^a	0.02	(0.01)	0.05	(0.02)	0.01	(0.01)	*
Number of observations	42	28	Ç	91	3	37	

^a The fraction of the corresponding group is shown.

b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%,

^{***}significant at 1%.

Table A4. Descriptive statistics – Chile

	A	11	NE	ET	No N	Sig.	
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Mean test d
Age	19.4	(0.05)	20.8	(0.11)	19.2	(0.06)	***
Male ^a	0.48	(0.01)	0.34	(0.02)	0.50	(0.01)	***
Years of education	11.3	(0.05)	11.5	(0.11)	11.3	(0.05)	**
Teenage parenthood ^a	0.12	(0.01)	0.31	(0.02)	0.08	(0.01)	***
Household monthly income per capita (log) ^b	274.4	(4.13)	217.6	(7.20)	283.9	(4.64)	***
Number of household members younger than 5 years old	0.29	(0.01)	0.50	(0.04)	0.25	(0.01)	***
Number of household members older than 65 years old	0.22	(0.01)	0.18	(0.02)	0.23	(0.01)	*
Numeracy skills (z-score) ^c	0.04	(0.02)	-0.07	(0.05)	0.06	(0.02)	**
Literacy skills (z-score) ^c	0.02	(0.02)	-0.16	(0.05)	0.06	(0.02)	***
Core self evaluation (z-score) ^c	0.03	(0.02)	-0.18	(0.05)	0.06	(0.02)	***
Grit test (z-score) ^c	0.03	(0.02)	-0.02	(0.05)	0.04	(0.02)	
Big Five							
Extraversion (z-score) ^c	0.01	(0.02)	-0.11	(0.05)	0.04	(0.02)	***
Agreeableness (z-score) ^c	-0.01	(0.02)	-0.01	(0.05)	0.00	(0.02)	
Conscientiousness (z-score) ^c	0.04	(0.02)	0.10	(0.05)	0.02	(0.02)	
Neuroticism (z-score) ^c	0.00	(0.02)	-0.02	(0.05)	0.00	(0.02)	
Openness (z-score) ^c	0.01	(0.02)	-0.04	(0.05)	0.02	(0.02)	
Education aspirations (additional years)	5.24	(0.05)	3.93	(0.12)	5.46	(0.06)	***
Expected return to tertiary education (log)	1.00	(0.01)	0.98	(0.03)	1.00	(0.01)	
Overestimates salary ^a	0.05	(0.00)	0.04	(0.01)	0.05	(0.00)	
Underestimates salary ^a	0.01	(0.00)	0.01	(0.00)	0.01	(0.00)	**
Activities							
Looking for a job ^a	0.11	(0.01)	0.39	(0.02)	0.07	(0.01)	***
Family caregiver ^a	0.36	(0.01)	0.54	(0.03)	0.33	(0.01)	***
Domestic work/family business ^a	0.73	(0.01)	0.81	(0.02)	0.71	(0.01)	***
Person with disability ^a	0.02	(0.00)	0.02	(0.01)	0.02	(0.00)	
Number of observations	27	38	39)2	23	46	

^a The fraction of the corresponding group is shown.

^b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%,

^{***}significant at 1%.

Table A5. Descriptive statistics - Colombia

	A	11	NI	EET	No N	Sig.	
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Mean test d
Age	19.2	(0.08)	20.0	(0.17)	19.0	(0.09)	***
Male ^a	0.48	(0.01)	0.34	(0.03)	0.51	(0.02)	***
Years of education	11.3	(0.07)	10.9	(0.17)	11.4	(0.08)	**
Teenage parenthood ^a	0.16	(0.01)	0.32	(0.03)	0.13	(0.01)	***
Household monthly income per capita (log) ^b Number of household members younger than 5 years old	325.8 0.27	(8.88) (0.02)	218.2 0.55	(14.84) (0.05)	346.0 0.21	(10.06) (0.01)	*** ***
Number of household members older than 65 years old	0.18	(0.01)	0.15	(0.03)	0.19	(0.01)	
Numeracy skills (z-score) ^c	0.00	(0.03)	-0.24	(0.07)	0.04	(0.03)	***
Literacy skills (z-score) ^c	0.02	(0.03)	-0.22	(0.08)	0.06	(0.03)	***
Core self evaluation (z-score) ^c	0.04	(0.03)	-0.22	(0.07)	0.09	(0.03)	***
Grit test (z-score) ^c	0.03	(0.03)	-0.06	(0.07)	0.04	(0.03)	
Big Five							
Extraversion (z-score) ^c	0.00	(0.03)	-0.12	(0.07)	0.03	(0.03)	*
Agreeableness (z-score) ^c	0.01	(0.03)	-0.09	(0.08)	0.03	(0.03)	
Conscientiousness (z-score) ^c	0.00	(0.03)	0.02	(0.07)	0.00	(0.03)	
Neuroticism (z-score) ^c	0.01	(0.03)	-0.06	(0.07)	0.02	(0.03)	
Openness (z-score) ^c	0.02	(0.03)	-0.03	(0.07)	0.03	(0.03)	
Education aspirations (additional years)	5.16	(0.07)	4.63	(0.15)	5.26	(0.08)	***
Expected return to tertiary education (log)	1.07	(0.02)	1.04	(0.04)	1.07	(0.02)	
Overestimates salary ^a	0.16	(0.01)	0.15	(0.02)	0.16	(0.01)	
Underestimates salary ^a	0.03	(0.00)	0.01	(0.01)	0.03	(0.01)	
Activities							
Looking for a job ^a	0.21	(0.01)	0.58	(0.03)	0.14	(0.01)	***
Family caregiver ^a	0.39	(0.01)	0.51	(0.03)	0.37	(0.01)	***
Domestic work/family business ^a	0.90	(0.01)	0.92	(0.02)	0.90	(0.01)	
Person with disability ^a	0.01	(0.00)	0.01	(0.01)	0.01	(0.00)	
Number of observations	13	14	2	08	11	106	

^a The fraction of the corresponding group is shown.

^b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%,

^{***}significant at 1%.

Table A6. Descriptive statistics – El Salvador

Variables	A	.11	NE	EET	No N	No NEET		
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Mean test d	
Age	19.4	(0.10)	20.5	(0.19)	19.2	(0.12)	***	
Male ^a	0.51	(0.02)	0.41	(0.04)	0.53	(0.02)	**	
Years of education	11.3	(0.10)	10.8	(0.19)	11.4	(0.12)	***	
Teenage parenthood ^a	0.09	(0.01)	0.15	(0.03)	0.07	(0.01)	**	
Household monthly income per capita (log)b	229.5	(7.94)	192.5	(13.31	238.6	(9.30)	***	
Number of household members younger than 5 years old	0.25	(0.02)	0.38	(0.05)	0.22	(0.02)	***	
Number of household members older than 65 years old	0.22	(0.02)	0.24	(0.05)	0.22	(0.02)		
Numeracy skills (z-score) ^c	0.01	(0.04)	-0.19	(0.08)	0.06	(0.04)	***	
Literacy skills (z-score) ^c	0.07	(0.04)	-0.21	(0.09)	0.14	(0.04)	***	
Core self evaluation (z-score) ^c	0.12	(0.04)	-0.05	(0.08)	0.16	(0.04)	**	
Grit test (z-score) ^c	0.09	(0.04)	-0.09	(0.09)	0.13	(0.04)	**	
Big Five								
Extraversion (z-score) ^c	0.05	(0.04)	-0.11	(0.09)	0.09	(0.04)	*	
Agreeableness (z-score) ^c	0.10	(0.04)	0.03	(0.08)	0.11	(0.04)		
Conscientiousness (z-score) ^c	0.01	(0.04)	0.10	(0.08)	-0.01	(0.04)		
Neuroticism (z-score) ^c	0.11	(0.04)	0.16	(0.09)	0.10	(0.04)		
Openness (z-score) ^c	0.06	(0.04)	0.07	(0.09)	0.06	(0.04)		
Education aspirations (additional years)	5.31	(0.09)	5.15	(0.19)	5.35	(0.11)		
Expected return to tertiary education (log)	0.81	(0.01)	0.79	(0.03)	0.81	(0.02)		
Overestimates salary ^a	0.03	(0.01)	0.03	(0.01)	0.03	(0.01)		
Underestimates salary ^a	0.06	(0.01)	0.06	(0.02)	0.05	(0.01)		
Activities								
Looking for a job ^a	0.19	(0.01)	0.57	(0.04)	0.10	(0.01)	***	
Family caregiver ^a	0.35	(0.02)	0.52	(0.04)	0.31	(0.02)	***	
Domestic work/family business ^a	0.89	(0.01)	0.94	(0.02)	0.87	(0.01)	***	
Person with disability ^a	0.01	(0.00)	0.01	(0.01)	0.01	(0.01)		
Number of observations	68	37	13	36	55	51		

^a The fraction of the corresponding group is shown.

^b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%,

^{***}significant at 1%.

Table A7. Descriptive statistics – Haiti

	A	All NE		ET	No N	No NEET	
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Mean test d
Age	19.9	(0.13)	21.7	(0.24)	19.5	(0.14)	***
Male ^a	0.46	(0.02)	0.30	(0.05)	0.49	(0.03)	***
Years of education	10.4	(0.15)	10.7	(0.33)	10.4	(0.16)	
Teenage parenthood ^a	0.03	(0.01)	0.09	(0.03)	0.02	(0.01)	**
Household monthly income per capita (log) ^b Number of household members younger than 5 years old	45.2 0.31	(3.10) (0.03)	39.6 0.39	(7.01) (0.07)	46.5 0.29	(3.46) (0.03)	
Number of household members older than 65 years old	0.29	(0.03)	0.25	(0.06)	0.30	(0.03)	
Numeracy skills (z-score) ^c	-0.09	(0.05)	-0.25	(0.11)	-0.05	(0.05)	*
Literacy skills (z-score) ^c	0.14	(0.04)	0.02	(0.10)	0.17	(0.05)	
Core self evaluation (z-score) ^c	-0.02	(0.04)	-0.02	(0.10)	-0.01	(0.05)	
Grit test (z-score) ^c	0.07	(0.05)	-0.03	(0.12)	0.09	(0.05)	
Big Five							
Extraversion (z-score) ^c	0.02	(0.04)	-0.10	(0.10)	0.05	(0.05)	
Agreeableness (z-score) ^c	-0.04	(0.04)	-0.15	(0.10)	-0.02	(0.05)	
Conscientiousness (z-score) ^c	-0.02	(0.04)	0.10	(0.09)	-0.05	(0.05)	
Neuroticism (z-score) ^c	0.02	(0.04)	-0.28	(0.11)	0.09	(0.05)	***
Openness (z-score) ^c	-0.04	(0.05)	-0.18	(0.10)	-0.01	(0.05)	
Education aspirations (additional years)	5.31	(0.14)	4.95	(0.33)	5.39	(0.16)	
Expected return to tertiary education (log)	1.09	(0.04)	1.11	(0.08)	1.08	(0.05)	
Overestimates salary ^a	0.04	(0.01)	0.02	(0.02)	0.04	(0.01)	
Underestimates salary ^a	0.04	(0.01)	0.03	(0.02)	0.04	(0.01)	
Activities							
Looking for a job ^a	0.20	(0.02)	0.41	(0.05)	0.15	(0.02)	***
Family caregiver ^a	0.65	(0.02)	0.70	(0.05)	0.64	(0.02)	
Domestic work/family business ^a	0.85	(0.02)	0.92	(0.03)	0.84	(0.02)	**
Person with disability ^a	0.09	(0.01)	0.08	(0.03)	0.09	(0.01)	
Number of observations	4	79	8	8	3'	91	

Note: Standard errors in parentheses.

a The fraction of the corresponding group is shown.

b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%,

^{***}significant at 1%.

Table A8. Descriptive statistics – Mexico

	A	All		NEET		No NEET	
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Mean test d
Age	19.1	(0.07)	20.2	(0.15)	18.8	(0.08)	***
Male ^a	0.45	(0.01)	0.23	(0.02)	0.51	(0.01)	***
Years of education	10.9	(0.07)	10.0	(0.15)	11.1	(0.08)	***
Teenage parenthood ^a	0.17	(0.01)	0.39	(0.03)	0.11	(0.01)	***
Household monthly income per capita (log) ^b	188.1	(5.69)	164.5	(13.86)	194.7	(6.16)	**
Number of household members younger than 5 years old	0.30	(0.02)	0.63	(0.04)	0.21	(0.01)	***
Number of household members older than 65 years old	0.10	(0.01)	0.10	(0.02)	0.10	(0.01)	
Numeracy skills (z-score) ^c	0.05	(0.03)	-0.19	(0.06)	0.11	(0.03)	***
Literacy skills (z-score) ^c	0.07	(0.02)	-0.08	(0.06)	0.11	(0.03)	***
Core self evaluation (z-score) ^c	0.04	(0.03)	-0.17	(0.06)	0.10	(0.03)	***
Grit test (z-score) ^c	0.05	(0.03)	-0.05	(0.05)	0.08	(0.03)	**
Big Five							
Extraversion (z-score) ^c	-0.02	(0.03)	-0.17	(0.06)	0.02	(0.03)	***
Agreeableness (z-score) ^c	0.00	(0.03)	-0.10	(0.06)	0.03	(0.03)	**
Conscientiousness (z-score) ^c	0.04	(0.02)	0.00	(0.05)	0.05	(0.03)	
Neuroticism (z-score) ^c	0.06	(0.03)	0.01	(0.05)	0.07	(0.03)	
Openness (z-score) ^c	0.05	(0.03)	-0.02	(0.05)	0.07	(0.03)	
Education aspirations (additional years)	5.77	(0.07)	5.64	(0.18)	5.80	(0.08)	
Expected return to tertiary education (log)	0.98	(0.02)	0.96	(0.03)	0.98	(0.02)	
Overestimates salary ^a	0.05	(0.01)	0.02	(0.01)	0.05	(0.01)	***
Underestimates salary ^a	0.37	(0.01)	0.43	(0.03)	0.35	(0.01)	***
Activities							
Looking for a job ^a	0.08	(0.01)	0.19	(0.02)	0.04	(0.01)	***
Family caregiver ^a	0.54	(0.01)	0.73	(0.02)	0.49	(0.01)	***
Domestic work/family business ^a	0.92	(0.01)	0.96	(0.01)	0.91	(0.01)	***
Person with disability ^a	0.03	(0.01)	0.02	(0.01)	0.03	(0.01)	
Number of observations	14	198	3	24	11	74	

^a The fraction of the corresponding group is shown.

^b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%,

^{***}significant at 1%.

Table A9. Descriptive statistics – Paraguay

		11	NEET		No NEET		Sig.
Variables	Mean	(SE)	Mean	(SE)	Mean	(SE)	Mean test ^d
Age	19.1	(0.08)	20.0	(0.15)	18.9	(0.09)	***
Male ^a	0.44	(0.01)	0.34	(0.03)	0.46	(0.02)	***
Years of education	11.1	(0.07)	10.8	(0.16)	11.2	(0.07)	**
Teenage parenthood ^a	0.12	(0.01)	0.29	(0.03)	0.08	(0.01)	***
Household monthly income per capita (log) ^b Number of household members younger than 5 years old	217.0 0.47	(6.31) (0.02)	181.5 0.72	(13.63) (0.06)	225.6 0.41	(7.08) (0.02)	*** ***
Number of household members older than 65 years old	0.22	(0.01)	0.19	(0.03)	0.23	(0.02)	
Numeracy skills (z-score) ^c	0.04	(0.03)	-0.21	(0.07)	0.10	(0.03)	***
Literacy skills (z-score) ^c	0.07	(0.03)	-0.16	(0.06)	0.13	(0.03)	***
Core self evaluation (z-score) ^c	0.04	(0.03)	-0.15	(0.06)	0.09	(0.03)	***
Grit test (z-score) ^c Big Five	0.01	(0.03)	-0.10	(0.06)	0.04	(0.03)	**
Extraversion (z-score) ^c	0.02	(0.03)	0.05	(0.06)	0.01	(0.03)	
Agreeableness (z-score) ^c	-0.01	(0.03)	-0.05	(0.07)	-0.01	(0.03)	
Conscientiousness (z-score) ^c	0.03	(0.03)	-0.04	(0.07)	0.05	(0.03)	
Neuroticism (z-score) ^c	0.00	(0.03)	-0.04	(0.06)	0.00	(0.03)	
Openness (z-score) ^c	0.03	(0.03)	-0.12	(0.06)	0.07	(0.03)	**
Education aspirations (additional years)	5.50	(0.07)	5.05	(0.15)	5.60	(0.08)	***
Expected return to tertiary education (log)	0.69	(0.02)	0.68	(0.03)	0.69	(0.02)	
Overestimates salary ^a	0.03	(0.01)	0.01	(0.01)	0.03	(0.01)	**
Underestimates salary ^a Activities	0.08	(0.01)	0.05	(0.01)	0.09	(0.01)	**
	0.16	(0.01)	0.38	(0.03)	0.11	(0.01)	***
Looking for a job ^a	0.10	(0.01)	0.56	(0.03)	0.11	(0.01) (0.02)	***
Family caregiver ^a	0.92	` /	0.96	` ′	0.94	(0.02) (0.01)	
Domestic work/family business ^a		(0.01)		(0.01)		` /	***
Person with disability ^a Number of observations	0.02	(0.00)	0.02	(0.01) 21	0.02	(0.00)	

^a The fraction of the corresponding group is shown.

^b Household monthly income per capita and expected higher education wage are deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^c Standardised measure.

^d Significance for mean tests between NEETs and non-NEETs: *significant at 10%, **significant at 5%,

^{***}significant at 1%.

Table A10. Marginal effects of Probit regressions: probability of being NEET - Women

	(1)	(2)	(3)	(4)	(5)
Age	0.03***	0.03***	0.03***	0.03***	0.03***
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Years of education	-0.02***	-0.01***	-0.01***	-0.02***	-0.02***
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Teenage parenthood=1	0.14***	0.14***	0.14***	0.13***	0.13***
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Household monthly income per capita (log) ^a	-0.04***	-0.04***	-0.04***	-0.03***	-0.03***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Number of household members below 5 years	0.03***	0.03***	0.03***	0.03***	0.03***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Number of household members above 65 years	-0.01	-0.01	-0.01	-0.01	-0.01
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Numeracy skills (z-score) ^b		-0.02***	-0.02***	-0.02***	-0.02***
		(0.00)	(0.00)	(0.00)	(0.00)
Literacy skills (z-score) ^b		-0.02***	-0.01***	-0.01**	-0.01**
		(0.00)	(0.00)	(0.00)	(0.00)
Core self evaluation (z-score) ^b			-0.02***	-0.02***	-0.02***
, ,			(0.00)	(0.00)	(0.00)
Grit test (z-score) ^b			-0.01	-0.01	-0.01
,			(0.00)	(0.00)	(0.00)
Big Five					
Extraversion (z-score) ^b			-0.01***	-0.01***	-0.01***
,			(0.00)	(0.00)	(0.00)
Agreeableness (z-score) ^b			-0.00	-0.00	-0.00
			(0.00)	(0.00)	(0.00)
Conscientiousness (z-score) ^b			-0.00	-0.00	-0.00
,			(0.00)	(0.00)	(0.00)
Neuroticism (z-score) ^b			0.00	0.00	0.00
,			(0.00)	(0.00)	(0.00)
Openness (z-score) ^b			-0.00	-0.00	-0.00
F			(0.00)	(0.00)	(0.00)
Education aspirations (additional years)				-0.02***	-0.02***
1				(0.00)	(0.00)
Expected return to tertiary education (log)					-0.00
					(0.01)
Overestimates salary=1					-0.02
•					(0.02)
Underestimates salary=1					0.02
•					(0.02)

Country dummies (base=Brazil)					
Chile	-0.04**	-0.05**	-0.05**	-0.06***	-0.06***
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Colombia	-0.02	-0.03	-0.03	-0.04*	-0.03
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
El Salvador	0.01	0.01	0.01	0.00	0.00
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Haiti	-0.10***	-0.10***	-0.09***	-0.10***	-0.10***
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Mexico	0.00	0.00	0.00	-0.00	-0.01
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Paraguay	-0.00	-0.00	-0.00	-0.01	-0.01
,	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Observations	4362	4362	4362	4362	4362
Pseudo R ²	0.11	0.12	0.12	0.13	0.13

Source: Millennials in LAC survey.

Note: Standard errors in parentheses. *Significant at 10%, ** significant at 5%, ***significant at 1%.

^a Household monthly income per capita and expected wages were deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b). b Standardized measure.

Table A11. Marginal effects of Probit regressions: probability of being NEET - Men

	(1)	(2)	(3)	(4)	(5)
Age	0.03***	0.03***	0.03***	0.03***	0.03***
O	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Years of education	-0.01***	-0.01***	-0.01***	-0.02***	-0.02***
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Teenage parenthood=1	0.13***	0.13***	0.13***	0.12***	0.12***
	(0.02)	(0.02)	(0.02)	(0.01)	(0.01)
Household monthly income per capita (log) ^a	-0.04***	-0.03***	-0.03***	-0.03***	-0.03***
, , , , ,	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Number of household members below 5 years	0.02***	0.02***	0.02***	0.02***	0.02***
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Number of household members above 65 years	-0.01	-0.01	-0.01	-0.01	-0.01
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Numeracy skills (z-score) ^b		-0.02***	-0.01***	-0.01***	-0.01***
		(0.00)	(0.00)	(0.00)	(0.00)
Literacy skills (z-score) ^b		-0.01***	-0.01***	-0.01**	-0.01**
		(0.00)	(0.00)	(0.00)	(0.00)
Core self evaluation (z-score) ^b			-0.02***	-0.02***	-0.02***
,			(0.00)	(0.00)	(0.00)
Grit test (z-score) ^b			-0.01	-0.00	-0.00
,			(0.00)	(0.00)	(0.00)
Big Five					
Extraversion (z-score) ^b			-0.01***	-0.01***	-0.01***
,			(0.00)	(0.00)	(0.00)
Agreeableness (z-score) ^b			-0.00	-0.00	-0.00
,			(0.00)	(0.00)	(0.00)
Conscientiousness (z-score) ^b			-0.00	-0.00	-0.00
` ,			(0.00)	(0.00)	(0.00)
Neuroticism (z-score) ^b			0.00	0.00	0.00
,			(0.00)	(0.00)	(0.00)
Openness (z-score) ^b			-0.00	-0.00	-0.00
,			(0.00)	(0.00)	(0.00)
Education aspirations (additional years)				-0.01***	-0.01***
				(0.00)	(0.00)
Expected return to tertiary education (log)					-0.00
					(0.01)
Overestimates salary=1					-0.02
					(0.02)
Underestimates salary=1					0.02
					(0.01)

Country dummies (base=Brazil)					
Chile	-0.04**	-0.04**	-0.04**	-0.05***	-0.05***
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Colombia	-0.02	-0.02	-0.02	-0.03	-0.03
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
El Salvador	0.01	0.01	0.01	0.00	0.00
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Haiti	-0.09***	-0.08***	-0.08***	-0.09***	-0.09***
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Mexico	0.00	0.00	0.00	-0.00	-0.01
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Paraguay	-0.00	-0.00	-0.00	-0.01	-0.01
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
Observations	3914	3914	3914	3914	3914
Pseudo R ²	0.11	0.12	0.12	0.13	0.13

Source: Millennials in LAC survey.

Note: Standard errors in parentheses. *Significant at 10%, ** significant at 5%, ***significant at 1%.

^a Household monthly income per capita and expected wages were deflated by each country's CPI and converted to 2011 PPP US dollars (World Bank, 2019b).

^b Standardized measure.